Author's response to reviews

Title: Selective repression of RET proto-oncogene in Medullary Thyroid Carcinoma by a natural alkaloid berberine

Authors:

Vishnu Muthuraj Kumarasamy Mr (vishnuk@email.arizona.edu)
Yoon-Joo Shin Ms (shin@pharmacy.arizona.edu)
John White Mr (muludnp@email.arizona.edu)
Daekyu Sun Dr (sun@pharmacy.arizona.edu)

Version: 2 Date: 19 December 2014

Author's response to reviews:

The following changes have been made to the manuscript main text:
1. The line and page numbers have been included in the main text.
2. The list of abbreviations have been included preceding the competing interest and author's contribution.